


Present: Alexei, Flemming, Gene, Tonko, Tommy, Prashanth, Yuri, Jim, and myself



== Hardware

	 — Switched to nitrogen flow in TPC

	 — preparing to regen oxygen and water scrubber

	 — preparing high-resolution CCD cameras for 3D sensors 

	 — east pole tip is already open (west, not yet)

	 — waiting on high radiation level to normalize before work on west side

	 — would be interesting to see actual readings in areas of hall

	 — question about spare sector modules in clean room (dispose of them?)

	 — Tonko mentions west side is now powered up

	 	 — also sector 24 FEEs (inner and outer) may come up and down as Tonko is working

	 	 — mentioning for information only so there is no surprise if power comes up and down



== Software

	 — Yunshan working on pp500 space charge for this year

	 	 — will be uploaded today or tomorrow

	 	 — starting point for next phase of calibration

	 — Shorted ring location (IWFC) needs to be localized before start of that phase

	 — Also need T0 nailed finalized

	 — Alexei will also try again to find the short on the hardware itself

	 — 7.7 GeV dE/dx still in progress 

	 — Yue-hang will work on fixed target 3.85 GeV space charge

	 	 — hope to start next week	 



== AOB

	 — Flemming has most material needed for talk next week

	 	 — will send around tomorrow for comments


